Author’s response to reviews

Title: Intraoperative Optical Coherence Tomography Assisted Analysis of Pars Plana Vitrectomy for Retinal Detachment in Morning Glory Syndrome: a case report

Authors:

Lyubomyr Lytvynchuk (lubko_l@yahoo.co.uk)
Carl Glittenberg (webmaster@carlglittenberg.com)
Siamak Ansari-Shahrezaei (ansari@retinacentervienna.com)
Susanne Binder (susanne@susannebinder.com)

Version: 2 Date: 23 Jul 2017

Author’s response to reviews:

A point-by-point detailed response to each reviewer/editorial point raised

BOPH-D-16-00409R1

1. Please consider changing the article type of your manuscript to "Case Report", as we feel this better reflects the contents of your study. Please amend your manuscript according to our guidelines for Case Report articles - https://bmcophthalmol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/case-report

Response: The authors changed the title of the article using description “a case report”.

2. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.
Abbreviations
Ethics approval and consent to participate
Consent for publication
Availability of data and material
Competing interests
Funding
Authors' contributions
Acknowledgements

Please see here for details on the information to be included in these sections:

https://bmcophthalmol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/case-report

If the information required is already provided in the main manuscript, please also copy the relevant statements to the Declarations.

If any of the sections are not relevant to your manuscript, please include the heading and write 'Not applicable' for that section.

Response: The Manuscript includes all required sections.

3. Please note that we require consent for publication of any potentially-identifying information that may be present in your manuscript, as well as for all case reports.
If your manuscript contains any individual person’s data in any form (including individual details, images or videos), consent to publish must be obtained from that person, or in the case of children, their parent or legal guardian. Please provide details of this consent in this section.

Please see our editorial policies for more information - https://www.biomedcentral.com/getpublished/editorial-policies#consent+for+publication

Response: A signed consent form is uploaded as a separate file.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: The authors uploaded all files as required.